Author's response to reviews

Title: MicroRNA-146a expresses in interleukin-17 producing T cells in rheumatoid arthritis patients

Authors:

Takuya Niimoto (tniimoto@yahoo.co.jp)
Tomoyuki Nakasa (tnakasa@smn.enjoy.ne.jp)
Masakazu Ishikawa (goemon1217@gmail.com)
Atsushi Okuhara (atshushiooku@yahoo.co.jp)
Bunichiro Izumi (b-izumi730@jk9.so-net.ne.jp)
Masataka Deie (snm3@hiroshima-u.ac.jp)
Osami Suzuki (osami@hiroshima-u.ac.jp)
Nobuo Adachi (nadachi@hiroshima-u.ac.jp)
Mitsuo Ochi (ochim@hiroshima-u.ac.jp)

Version: 2 Date: 6 June 2010

Author's response to reviews: see over
June, 06, 2010

Dr. Melissa Norton, MD
BMC Musculoskeletal Disorders Editor in chief
Dear Dr. Melissa Norton

Re: “MicroRNA-146a expresses in interleukin-17 producing T cells in rheumatoid arthritis patients” by Takuya Niimoto, Tomoyuki Nakasa.
(submitted on May 24 2010. MS: 4970360233971936)

Thank you indeed for reporting for this time.
We submitted my microarray date to ArrayExpress (June, 04, 2010).
However, I do not yet receive accession number. I want you to wait a little more if possible.
In addition, our array date was added figure.
We thank you for considering our paper for publication in BMC Musculoskeletal Disorders and look forward to hearing from you at your earliest convenience.

Sincerely yours,
Takuya Niimoto, MD
Tomoyuki Nakasa, MD, PhD
Department of Orthopaedics Surgery, Programs for Applied Biomedicine, Division of Clinical Medical Science, Graduate School of Biomedical Science
1-2-3 Kasumi, Minami-ku, Hiroshima, 734-8551, Japan
Tel: +81-82-257-5233
Fax: +81-82-257-5234
Takuya Niimoto: tniimoto@yahoo.co.jp
Tomoyuki Nakasa: tnakasa@smn.enjoy.ne.jp